Author's response to reviews

Title: Islet Expression of the DNA Repair Enzyme 8-OxoG DNA glycosylase (Ogg1) in Human Type 2 Diabetes

Authors:

Bjorn Tyrberg (btyrberg@ucsd.edu)
Kamen A Anachkov (k_anachkov@doctor.com)
Sergio A Dib (sadib@endocrino.epm.br)
Jessica Wang-Rodriguez (jwrodriguez@ucsd.edu)
Kun-Ho Yoon (yoonk@cmc.cuk.ac.kr)
Fred Levine (flevine@ucsd.edu)

Version: 5 Date: 24 Apr 2002

The Figure legends are now changed to #1-6 as indicated by the editor.